

Responses to the reviewer's comments

Original comment:

Thank you for giving me the opportunity to review this Manuscript. The authors have reported a rare complication of colonoscopic , which will be useful to report literature. The manuscript is very well written without any major shortcoming. I would like to congratulate the authors for quality manuscript.

Response:

Dear Reviewer,

Thank you for your thorough review and valuable feedback on our manuscript. We greatly appreciate your positive comments and your congratulations on the quality of our work.

We have noted your observation regarding the language quality of the manuscript, specifically your identification of the need for minor language polishing (Grade B). We have taken immediate steps to address this by engaging a professional language editing service to refine and polish the text thoroughly. This has ensured that the language in our manuscript now meets the high standards of academic writing and clarity expected by the journal.

We hope that the revised manuscript meets your expectations and those of the journal. We are enthusiastic about the potential publication of our work, which we believe will contribute significantly to the literature on gastrointestinal endoscopic complications.

Thank you once again for your constructive comments and for your part in improving our manuscript.

Yours sincerely,

Saiheng Xiang & Guoqiang Xu